Author’s response to reviews

Title: Influence of oxygen tension on myocardial performance. Evaluation by tissue Doppler imaging

Authors:

Ole Frobert (frobert@dadlnet.dk)
Jacob Moesgaard (jm@doktor.dk)
Egon Toft (ET@healthntech.dk)
Steen H Poulsen (steen.hvitfeldt@dadlnet.dk)
Peter Sogaard (psogaard@dadlnet.dk)

Version: 7 Date: 2 November 2004

Author’s response to reviews:

Running title and author qualifications on the title page have been removed. The figure legends are now after the reference list.